**Author's response to reviews**

**Title:** Television viewing through ages 2-5 years and bullying involvement in early elementary school

**Authors:**
- Marina Verlinden (m.verlinden@erasusmc.nl)
- Henning W. Tiemeier (h.tiemeier@erasusmc.nl)
- René Veenstra (d.r.veenstra@rug.nl)
- Cathelijne L. Mieloo (cl.mieloo1@rotterdam.nl)
- Wilma Jansen (w.jansen@rotterdam.nl)
- Vincent W.V. Jaddoe (v.jaddoe@erasusmc.nl)
- Hein Raat (h.raat@erasusmc.nl)
- Albert Hofman (a.hofman@erasusmc.nl)
- Frank C. Verhulst (f.verhulst@erasusmc.nl)
- Pauline W. Jansen (p.w.jansen@erasusmc.nl)

**Version:** 4  
**Date:** 4 February 2014

**Author's response to reviews:**

To: Editor BMC Public Health  
Date: 03-02-2014  
Concerning: manuscript re-submission

**Dear Editor,**

Please find enclosed the revised version of our manuscript “Television viewing through ages 2-5 years and bullying involvement in early elementary school”.

We were very pleased that the reviewers now evaluate our paper as being ready to be published provided we undergo the minor revision as described in the reviewer’s report. We have adapted our manuscript following all the reviewer’s suggestions. Our replies to the comments of the reviewer are presented in the attached document. For convenience and ease of reading, we highlighted the changes in the text of the manuscript (in grey color). We would like to thank you and all the reviewers for the multiple readings of our manuscript that helped us improve the quality of the manuscript.

**We look forward to your evaluation.**

**With kindest regards,**  
Pauline Jansen